Author's response to reviews

Title: Mucosal atrophy in collagenous colitis: A case report

Authors:

Michael Majores (mmajores@uni-bonn.de)
Steffen Wunsch (familie-wunsch@web.de)
Bernd Blume (dr.b.blume@t-online.de)
Hans-Peter Fischer (hans-peter.fischer@ukb-uni-bonn.de)
Christoph Reichel (christoph.reichel@web.de)

Version: 3 Date: 9 June 2011

Author's response to reviews: see over
Dear Mr. Shipley,

Please find enclosed the re-revised version of our manuscript entitled „Mucosal atrophy in collagenous colitis: A case report.” in which all of the reviewers’ suggestions have been addressed. A detailed list of changes is enclosed.

We hope that the re-revised manuscript meets your and the referees’ expectations and is now suitable for publication. In addition, we want to express our gratitude to the reviewers for their fair and very constructive suggestions, which improved the quality of our manuscript considerably.

Thank you very much for your time and consideration.

Sincerely yours

Christoph Reichel
Point-to-point consideration of the reviewers' suggestions:

Reviewer Magnus Ström:
The Figure number has been checked.

Reviewer Unji Umeno:
The author should show the endoscopic picture of normal appearance of the cecum in 1998.

We do now show the endoscopic picture of the cecum in 1998 as insert in the new Figure 1A.